Author's response to reviews

Title: Psychometric properties of Greek versions of the Modified Corah Dental Anxiety Scale (MDAS) and the Dental Fear Survey (DFS)

Authors:

Trilby Coolidge (tcoolidg@u.washington.edu)
Konstantinos N Arapostathis (koarap@dent.auth.gr)
Dimitris Emmanouil (emmand@dent.uoa.gr)
Nikolaos Dabarakis (nikosd@dent.auth.gr)
Antonis Patrikiou (tonypatr@yahoo.com)
Nikolaos Economides (econom@dent.auth.gr)
Nikolaos Kotsanos (kotsanos@dent.auth.gr)

Version: 7 Date: 24 September 2008

Author's response to reviews:

Small grammatical change on p. 12, from "Thus, it appears that the Greek versions these measures appear to operate..." to "Thus, it appears that the Greek versions of these measures operate..."